Author’s response to reviews

Title: Emergence of A New Crimean-Congo Hemorrhagic Fever Virus Strain in Turkey

Authors:

Kenan Midilli (kmidilli@gmail.com)
Aysen Gargili (agargili@yahoo.com)
Onder Ergonul (onderergonul@yahoo.com)
Murat Elevli (melevli@gmail.com)
Sevgi Ergin (sergin@yahoo.com)
Nesrin Turan (nturan@gmail.com)
Gönül Sengoz (drgonul@hasekihastanesi.gov.tr)
Recep Ozturk (ozturkrecep@superonline.com)
Mehmet Bakar (mbakar@gmail.com)

Version: 2 Date: 7 July 2008

Author’s response to reviews:

1. Abstract was revised.
2. Contributions of the authors were written.
3. In methods section the sentence, "This was an observational study, which was conducted in compliance with Helsinki declaration" was inserted.
4. "No competing interest for authors" sentence was added.